OODS Tutorial 8

© 2000/2001 Michael Golm Universität Erlangen Nürnberg, Institut für Informatik 4

Homework Assignment #6: Remote Method Invocation 0 points

The purpose of this assignment is to learn using threads and the RMI package. The directory/proj/i4oods/pub/assignment6 contains the following files:

DrawingApplet.html: Used to start DrawingApplet.class
DrawingApplet.java: Applet to draw lines
Line.java: Represents a line
LineSet.java: Container for Line objects
LineSetObserver.java: Interfacethat must be implementedby an object that wants to be in-

formed about changes of the LineSet. These files implement a non-distributed line editor.

a) Modifythese filesto create a distributedapplicationsimilarto assignment #5, but use Remote MethodInvocation(RMI)instead ofsockets forcommunication.The LineSet should run on the Web server as a remote object. The DrawingApplets should obtain a reference to the remote LineSet object and use the remote object instead ofthe localone. Thismeans that allapplets should use the same LineSet object. Whenever new lines are added to the line set, allapplets shoulddrawthe new lines.Thereforeyouhave to implementthe LineSetOb-

server as a remote object that runs in the applet’s JVM.

The LineSet objectis used bymore than one user at the same time.Thinkabout the consequences for the consistency of this object and change the LineSet class appropriately.

• Modify as less as possible of the original files.

• Start the registry with your user ID as the port number.

• Ifexceptions occurthat can notbe handled ina suitablemanner printoutan error message and the stack trace.

How to start and test your RMI server in the CIP computer room:

• There is one special machine oods.informatik.uni-erlangen.de(whichis another name for faui01a) which runs the Web server for Assignment #6.

• Each OODS member has a directoryin /proj/i4oods/wwwthat is also available through this Web server via the URLhttp://oods.informatik.uni-erlangen.de/.Youshould put yourAppletand the RMI stubs intothis directoryor create sub-directories- remember that the directorieshave to reflectyour package structure. The wholewwwdirectorytree and each filetherein has to be readable forevery user because otherwise the Web server willnot be able to export the filesand you willget an error message.

• Youwillalso have to start the RMIregistryand your server on this machine. Don’tforgetto set the codebase to the URLpointingto the top package in your wwwtree. The reason forboth actions is that applets are only allowed to connect back to the host that they were loaded from.